Author's response to reviews

Title: Quality of life, mental health and health beliefs in haemodialysis and peritoneal dialysis patients: Investigating differences in early and later years of current treatment.

Authors:

Maria Ginieri- Coccossis (margkok@med.uoa.gr)
Paraskeyi Theofilou (theofi@otenet.gr)
Claire Synodinou (clarysyn@panteion.gr)
Constantinos Soldatos (egslelabath@hol.gr)

Version: 3 Date: 9 October 2008

Author's response to reviews:

The major revisions have been addressed. In the abstract, the word 'Objective' has been changed to 'Introduction'

The footnotes have been removed and instead incorporated in the text.

The minor revisions have also been addressed.

The 'Authors' contributions' is now formatted into a paragraph.

The 'Abbreviations' has been formatted into a paragraph. A statement has been added that all authors read and approved the final manuscript.